Author's response to reviews

Title: Case report: Type I Gaucher disease with exophthalmos and pulmonary arteriovenous malformation

Authors:

Chun-An Chen (chenca@ntu.edu.tw)
Nelson LS Tang (nelsontang@cuhk.edu.hk)
Yin-Hsiu Chien (chienyh@ha.mc.ntu.edu.tw)
Wei-Min Zhang (nelsontang@cuhk.edu.hk)
Jou-Kou Wang (jkw@ha.mc.ntu.edu.tw)
Wuh-Liang Hwu (hwu@ha.mc.ntu.edu.tw)

Version: 3 Date: 12 April 2005

Author's response to reviews:

Dear reviewer:
We really appreciate your comments on this article. The following are our replies to your suggestions. We have also revised our manuscript accordingly.

Comment 1: Eliminate the statement "this case......for the mechanism of Gaucher disease".
A: We agree with your suggestion and have deleted similar sentences in the abstract and conclusion.
Comment 2: Should not skew the discussion to hepatopulmonary syndrome.
A: We have modified our statement so that hepatopulmonary syndrome would no longer be our explanation for her pulmonary AVM.
Comment 3: Send to a native speaker for improving the grammar and spelling.
A: This manuscript has been edited by a native speaker.